Author’s response to reviews

Title: Occipital Peripheral Nerve Stimulation in the management of chronic intractable occipital neuralgia in a patient with neurofibromatosis type-1.

Authors:

Ioannis M Skaribas (iskaribas@choosegha.com)
Octavio Calvillo (octaviocalvillo@yahoo.com)

Version: 2 Date: 20 February 2010

Author’s response to reviews:

This is the First report of its kind in the medical literature. Specifically the usage of Ultrasound to guide stimulation electrode placement for OPNS in a patient with both Neurofibromatosis and Occipital neuralgia has not been previously reported. This original case report will have a clinical impact not only for the interventional Anesthesiologist practicing Pain medicine but will also provide an alternative mode of therapy for Neurologists and Primary care Doctors treating these patients.